Author’s response to reviews

Title: Risk factors for failure of revision total hip arthroplasty using a Kerboull-type acetabular reinforcement device

Authors:

Shinya Hayashi (s11793290@yahoo.co.jp)

Takayuki Nishiyama (takakyon2@aol.com)

Shingo Hashimoto (shingo1023@theia.ocn.ne.jp)

Tomoyuki Matsumoto (matsun@m4.dion.ne.jp)

Koji Takayama (kojitakayama1978@gmail.com)

Kazunari Ishida (s11793290ishida@yahoo.co.jp)

Kotaro Nishida (s11793290nishida@yahoo.co.jp)

Ryosuke Kuroda (s11793290kuroda@yahoo.co.jp)

Version: 2 Date: 31 Jul 2017

Author’s response to reviews:

Answer to Editor comments: Editor Comments: Figures: Each figure has to be able to stay alone. Answer: Thank you for your suggestion. We separated each figure alone. Please describe abbreviations in the figures' caption. Answer: Thank you for your suggestion. We added abbreviations in the figures' caption. Figure 5: Please express the survival rate as percentage. Please delete the decimals (x-axis) as you show only full years. Answer: Thank you for your suggestion. We changed survival rate as percentage and delete the decimals (x-axis). Figure 1 b: It should be part of quality control to display the entire stem of the prosthesis. We understand that the main focus of the manuscript lies on the acetabular component. However, a better image may prevent the readers' suggestion that the stem was not evaluated sufficiently. We would like to leave the decision to exchange the image to the authors. Answer: Thank you for your suggestion. We changed X-rays according to your advice. Keywords: Please try to choose more meaningful keywords. Answer: Thank you for your suggestion. We changed keywords to revision total hip arthroplasty, KT plate, allograft, beta-tricalcium phosphate, hydroxyapatite.